Responses to Reviewer #1

Specific Comments to Authors: This is a well-designed study. The manuscript is well-written and concise. The results are consistent and of relevance. I would recommend to accept this manuscript for publication. There are some typographic errors. There are very long sentences. To improve readability, consider breaking this into multiple sentences. The authors are encouraged to proof-read thoroughly the text before resubmission. English must be excellent.

Responses: Thank you for your comments. We have proof-read the manuscript, corrected typographic errors and tried to break very long sentences into multiple sentences. In addition, Dr. Kaiying Zhang (Department of Anesthesiology, The University of Texas Health Science Center at Houston, USA) has helped us in English editing.

Responses to Reviewer #2

Specific Comments to Authors: This manuscript was well, concisely and coherently organized and presented. Therefore, it reaches the accept level to this journal.

Responses: Thank you for your comments.